Author's response to reviews

Title: One year outcomes of a mentoring scheme for female academics: a pilot study at the Institute of Psychiatry, King's College London

Authors:

- Rina Dutta (rina.2.dutta@kcl.ac.uk)
- Sarah L Hawkes (sarah.l.hawkes@hotmail.co.uk)
- Elizabeth Kuipers (elizabeth.kuipers@kcl.ac.uk)
- David Guest (david.guest@kcl.ac.uk)
- Nicola T Fear (nicola.t.fear@kcl.ac.uk)
- Amy C Iversen (amy.c.iversen@kcl.ac.uk)

Version: 6 Date: 29 March 2011

Author's response to reviews:

1) King's College London need to be the first part of the address for Dr. Amy Iversen

2) I have noticed an error with the mean differences and 95% confidence intervals presented in the first row of Table 1. These currently read as: 0.97 (-0.20 to 2.14) and 0.96 (-0.57 to 2.49). These should be replaced with: -0.97 (-2.14 to 0.20) and -0.96 (-2.49 to 0.57).